Author's response to reviews

Title: Health Insurance, Neighborhood Income, and Emergency Department Usage by Utah Children 1996-1998

Authors:

Anthony Suruda (tonysuruda@aol.com)
Thomas J Burns (tburns@ou.edu)
Stacey Knight (Stacey.Knight@hsc.utah.edu)
J. MICHAEL Dean (Mike.Dean@hsc.utah.edu)

Version: 6 Date: 17 March 2005

Author's response to reviews:

March 17, 2005

Dear Ms. Shaw,

I have made the formatting changes suggested in your e-mail of March 14, 2005. I have added a sentence to the Abstract Background to put our work in context. I reconfigured the Discussion section so that a Conclusions section follows.

I appreciate your patience in dealing with the formatting issues. If there are further changes needed please let me know.

Anthony Suruda MD MPH